Author’s response to reviews

Title: Investigating Usability of Mobile Health Applications in Bangladesh

Authors:

Muhammad Nazrul Islam (nazrul@cse.mist.ac.bd)
Md. Mahboob Karim (mahboob4146@yahoo.com)
Toki Tahmid Inan (tokiinan5@gmail.com)
A.K.M. Najmul Islam (najmul.islam@utu.fi)

Version: 3 Date: 19 Jan 2020

Author’s response to reviews:

Dear Dr. Editor-in-Chief,

We are pleased, that you gave us the possibility to revise this manuscript for one more time. We truly appreciate the Assistant Editor’s comments related to the manuscript format, including the abbreviations, ethical approval, consent for publications, availability of data and material, authors’ contributions and appendix. In revision, we have addressed each of these concerns.

We hope that these revisions improve the paper in such a way that you and the assistant editor now deem it worthy of publication in BMC Medical Informatics and Decision Making Journal.

Best regards,

Dr. Muhammad Nazrul Islam